Author's response to reviews

Title: Tai Chi-based exercise program provided via telerehabilitation compared to home visits in a post-stroke population who have returned home without intensive rehabilitation: a randomized, non-inferiority clinical trial

Authors:

Michel Tousignant Prof (michel.tousignant@usherbrooke.ca)
Hélène Corriveau Prof (helene.corriveau@usherbrooke.ca)
Dahlia Kairy Prof (dahlia.kairy@umontreal.ca)
Katherine Berg Prof (katherine.berg@utoronto.ca)
Marie-France Dubois Prof (marie-france.dubois@usherbrooke.ca)
Sylvie Gosselin Dr (sylvie.gosselin@usherbrooke.ca)
Richard H Swartz Prof (rick.swartz@sunnybrook.ca)
Jean-Martin Boulanger Dr (jbouboul@hotmail.com)
Cynthia Danells (cynthia.danells@sunnybrook.ca)

Version: 4 Date: 16 October 2013

Author's response to reviews:

We did the adjustments to the manuscript, to make it fits well to the journal style.